

Citation: Auld AF, Kerkhoff AD, Hanifa Y, Wood R, Charalambous S, Liu Y, et al. (2021) Derivation and external validation of a risk score for predicting HIV-associated tuberculosis to support case finding and preventive therapy scale-up: A cohort study. PLoS Med 18(9): e1003739. https://doi.org/ 10.1371/journal.pmed.1003739

Academic Editor: Sydney Rosen, Boston University School of Public Health, UNITED STATES

Received: November 24, 2020

Accepted: July 21, 2021

Published: September 7, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003739

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: The authors confirm that, for IRB-approved reasons, some access

RESEARCH ARTICLE

Derivation and external validation of a risk score for predicting HIV-associated tuberculosis to support case finding and preventive therapy scale-up: A cohort study

Andrew F. Auld¹*, Andrew D. Kerkhoff², Yasmeen Hanifa³, Robin Wood⁴, Salome Charalambous⁵, Yuliang Liu¹, Tefera Agizew⁶, Anikie Mathoma^{6,7}, Rosanna Boyd⁶, Anand Date¹, Ray W. Shiraishi¹, George Bicego¹, Unami Mathebula-Modongo⁶, Heather Alexander¹, Christopher Serumola⁶, Goabaone Rankgoane-Pono⁸, Pontsho Pono⁸, Alyssa Finlay⁶, James C. Shepherd^{6,9}, Tedd V. Ellerbrock⁹, Alison D. Grant^{3,10°}, Katherine Fielding^{3°}

1 Division of Global HIV & TB, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 2 Division of HIV, Infectious Diseases and Global Medicine at Zuckerberg San Francisco General Hospital and Trauma Center, Department of Medicine, University of California San Francisco, San Francisco, California, United States of America, 3 TB Centre, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 The Desmond Tutu HIV Centre, Institute of Infectious Disease and Molecular Medicine, Cape Town, South Africa, 5 Aurum Institute, Johannesburg, South Africa, 6 Division of TB Elimination, Centers for Disease Control and Prevention, Gaborone, Botswana, 7 University of Botswana, Gaborone, Botswana, 8 Ministry of Health and Wellness, Gaborone, Botswana, 9 Yale University School of Medicine, New Haven, Connecticut, United States of America, 10 Africa Health Research Institute, School of Laboratory Medicine and Medical Sciences, College of Health Sciences, University of KwaZulu-Natal, Durban, South Africa

These authors contributed equally to this work.
* aauld@cdc.gov

Abstract

Background

Among people living with HIV (PLHIV), more flexible and sensitive tuberculosis (TB) screening tools capable of detecting both symptomatic and subclinical active TB are needed to (1) reduce morbidity and mortality from undiagnosed TB; (2) facilitate scale-up of tuberculosis preventive therapy (TPT) while reducing inappropriate prescription of TPT to PLHIV with subclinical active TB; and (3) allow for differentiated HIV–TB care.

Methods and findings

We used Botswana XPRES trial data for adult HIV clinic enrollees collected during 2012 to 2015 to develop a parsimonious multivariable prognostic model for active prevalent TB using both logistic regression and random forest machine learning approaches. A clinical score was derived by rescaling final model coefficients. The clinical score was developed using southern Botswana XPRES data and its accuracy validated internally, using northern Botswana data, and externally using 3 diverse cohorts of antiretroviral therapy (ART)-naive and ART-experienced PLHIV enrolled in XPHACTOR, TB Fast Track (TBFT), and Gugulethu studies from South Africa (SA). Predictive accuracy of the clinical score was compared

restrictions apply to the data underlying the findings. Although the patient-level data do not include patient names, this IRB decision is in the interest of ensuring patient confidentiality. An individual may email the CDC Division of Global HIV & TB science office (gapmts@cdc.gov) to request the data.

Funding: This research has been supported by the President's Emergency Plan for AIDS Relief (PEPFAR) through the US Centers for Disease Control and Prevention. The funder had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: AG has received research grants (not related to this work) from the Medical Research Council, Economic and Social Research Council, Bill and Melinda Gates Foundation, National Institute of Allergy and Infectious Diseases, Wellcome Trust, Research England, and USAID.

Abbreviations: AIC, Akaike information criteria; ART, antiretroviral therapy: AUROC, area under the receiver operating characteristic; BIC, Bayesian information criteria; BMI, body mass index; CDC, Centers for Disease Control and Prevention; COVID-19, Coronavirus Disease 2019; CRP, Creactive protein; CRT, cluster randomized trial; CXR, chest radiography; EMR, electronic medical record; HRDC, Health Research and Development Committee; IRB, Institutional Review Board; LMIC, low- and middle-income country; MFP, multivariable fractional polynomial; NNS, number needed to screen; NPV, negative predictive value; PEPFAR. US President's Emergency Plan for AIDS Relief; PLHIV, people living with HIV; POC, pointof-care; PPV, positive predictive value; SA, South Africa; SOC, standard of care; SSA, sub-Saharan Africa; TB, tuberculosis; TBFT, TB Fast Track; TPT, tuberculosis preventive therapy; TRIPOD, Transparent Reporting of a multivariable prediction model for Individual Prognosis or Diagnosis; WHO, World Health Organization.

with the World Health Organization (WHO) 4-symptom TB screen. Among 5,418 XPRES enrollees, 2,771 were included in the derivation dataset; 67% were female, median age was 34 years, median CD4 was 240 cells/µL, 189 (7%) had undiagnosed prevalent TB, and characteristics were similar between internal derivation and validation datasets. Among XPHAC-TOR, TBFT, and Gugulethu cohorts, median CD4 was 400, 73, and 167 cells/µL, and prevalence of TB was 5%, 10%, and 18%, respectively. Factors predictive of TB in the derivation dataset and selected for the clinical score included male sex (1 point), ≥1 WHO TB symptom (7 points), smoking history (1 point), temperature >37.5°C (6 points), body mass index (BMI) <18.5kg/m² (2 points), and severe anemia (hemoglobin <8g/dL) (3 points). Sensitivity using WHO 4-symptom TB screen was 73%, 80%, 94%, and 94% in XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively, but increased to 88%, 87%, 97%, and 97%, when a clinical score of \geq 2 was used. Negative predictive value (NPV) also increased 1%, 0.3%, 1.6%, and 1.7% in XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively, when the clinical score of \geq 2 replaced WHO 4-symptom TB screen. Categorizing risk scores into low (<2), moderate (2 to 10), and high-risk categories (>10) yielded TB prevalence of 1%, 1%, 2%, and 6% in the lowest risk group and 33%, 22%, 26%, and 32% in the highest risk group for XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively. At clinical score \geq 2, the number needed to screen (NNS) ranged from 5.0 in Gugulethu to 11.0 in XPHACTOR. Limitations include that the risk score has not been validated in resource-rich settings and needs further evaluation and validation in contemporary cohorts in Africa and other resource-constrained settings.

Conclusions

The simple and feasible clinical score allowed for prioritization of sensitivity and NPV, which could facilitate reductions in mortality from undiagnosed TB and safer administration of TPT during proposed global scale-up efforts. Differentiation of risk by clinical score cutoff allows flexibility in designing differentiated HIV–TB care to maximize impact of available resources.

Author summary

Why was this study done?

- Tuberculosis (TB) remains the most common cause of death among people living with HIV (PLHIV) and is often undiagnosed at time of death.
- Rapid scale-up of tuberculosis preventive therapy (TPT) to 13 million PLHIV in lowand middle-income countries (LMICs) has been proposed for 2021; however, active TB is commonly asymptomatic and therefore missed by current WHO-recommended 4symptom TB screening rules.
- Therefore, more sensitive TB screening tools are needed to better facilitate early TB diagnosis and safer scale-up of TPT to PLHIV by avoiding TPT prescription to clients with asymptomatic active TB, who need TB treatment.

What did the researchers do and find?

- We derived a TB risk score for PLHIV from XPRES trial data and validated the score on 3 external datasets.
- We prioritized high sensitivity and ability to correctly rule out TB (i.e., high negative predictive value (NPV)) at key time points in care such as HIV clinic enrollment and before TPT prescription.
- Both logistic regression and random forest machine learning approaches were used to identify the 6 most important predictors, commonly available in LMIC clinic settings.
- In the external datasets, TB risk score ≥2 had higher sensitivity (87% to 97%) than WHO 4-symptom screening rule and increased NPV by 0.3% to 1.7%.
- Three risk groups were identified by the score, with active TB prevalence in external datasets ranging from 1% to 6% in the lowest to 22% to 32% in the highest risk groups.

What do these findings mean?

- Following further validation, this clinical score could improve early detection of active TB to reduce morbidity and mortality from undiagnosed TB.
- Use of the clinical score cutoff of ≥2 during the proposed TPT scale-up for 13 million PLHIV could potentially avoid many thousands of PLHIV with active TB being inappropriately prescribed TPT.
- By differentiating 3 risk groups, the score also allows for the development of differentiated service delivery models suitable for LMIC.

Introduction

Tuberculosis (TB) remains the most common cause of death among people living with HIV (PLHIV), with 251,000 HIV-associated TB deaths in 2018, over 95% of which occurred in lowand middle-income countries (LMICs) [1]. Among PLHIV who die from TB, TB often remains undiagnosed at the time of death [2,3]. The World Health Organization (WHO) recommends a 4-symptom TB screening rule (i.e., for cough, weight loss, night sweats, or fever) to determine which PLHIV need investigation for active TB and which are eligible for immediate tuberculosis preventive therapy (TPT) [4]. WHO 4-symptom TB screening rule is recommended for LMIC regardless of expected prevalence of active TB, setting (e.g., high or low TB incidence settings), or antiretroviral therapy (ART) status (ART-naive or ART-experienced) [4].

However, screening accuracy of WHO 4-symptom screening rule varies by population, setting, and ART status, raising the question whether a "one-size-fits-all" screening rule is appropriate. For example, a recent meta-analysis observed that while sensitivity of WHO 4-symptom TB screening rule is about 89% among ART-naive PLHIV, it is only 51% among people on ART due to a higher prevalence of subclinical TB (i.e., asymptomatic disease, caused by viable *Mycobacterium tuberculosis*, detectable by microbiologic tests or radiography [5]) among stable ART patients [6–8]. At a time when global health donors have committed to reaching over 13 million PLHIV on ART with TPT by 2021 [9], low sensitivity of WHO 4-symptom screening rule for active TB among PLHIV on ART warrants consideration of more sensitive screening approaches [10]. Although new WHO guidelines recommend adding chest radiography (CXR) to the screening rule for PLHIV on ART to increase sensitivity and negative predictive value (NPV), CXR is not available in many LMIC clinic settings; this comes at the expense of specificity, carries significant additional costs and operational challenges, and might hinder rather than expedite TPT scale-up in some LMIC settings [6,11]. Subclinical TB can also be present among severely immune compromised PLHIV [12] and among pre-ART patients without advanced disease in high prevalence settings [13], among whom failing to detect active TB can have serious health consequences for patients and impede disease control activities [14]. Finally, WHO 4-symptom screening rule does not allow TB risk differentiation into low-, moderate-, and high-risk groups, which might inform differentiated models of care.

Therefore, we aimed to develop a predictive clinical score based on variables commonly available in resource-constrained clinics, to define a range of cutoffs, with associated screening sensitivity, specificity, NPV, positive predictive value (PPV), percentage screened into diagnostic test algorithms, and number needed to screen (NNS) to detect one person with active TB.

Methods

We used data from the Xpert Package Rollout Evaluation using a stepped wedge design (XPRES) trial conducted in Botswana to derive the predictive TB clinical score [15,16]. We split XPRES cohort data geographically into 11 southern and 11 northern clinics to serve as an internal derivation and validation dataset, respectively. We used 2 different but complementary modeling approaches to generate a parsimonious TB clinical risk score comprised of variables easily available in a resource-constrained clinic setting: (1) logistic regression models; and (2) random forest machine learning models. Random forest machine learning models are particularly useful for identifying important nonlinear associations between predictors and outcomes because the modeling approach does not rely on assumptions of average linear or curvilinear associations [17]. Having derived the clinical score, we then used data from 3 other settings to externally validate the derived clinical score.

Firstly, we used prospective cohort data for XPHACTOR study enrollees from Gauteng Province, South Africa (SA) [18]. XPHACTOR trial enrollees differed from XPRES enrollees in that 89% of enrollees were stable on ART at study enrollment versus 0% at XPRES study enrollment. Secondly, we used cluster randomized trial (CRT) data from the TB Fast Track (TBFT) trial from Gauteng, Limpopo, and North West Provinces in SA. TBFT trial enrollees differed from XPRES enrollees in that all had advanced HIV disease (all had CD4 count <150 cells/µL at study enrollment), but, similar to XPRES, no TBFT trial enrollees had started taking ART [19]. Thirdly, we used prospective cohort data from the Western Cape, SA (the Gugulethu cohort). Gugulethu cohort enrollees differed from XPRES enrollees in that Gugulethu has a very high background TB notification rate in the general population (>1,000/100,000 population [20]) compared with XPRES enrollees in Botswana where background TB notification rates were \leq 500/100,000 [21], although similar to XPRES all Gugulethu cohort enrollees had not started ART at the time of study enrollment [22]. We compared screening accuracy of our derived clinical scores with existing WHO TB symptom screening criteria for active TB among PLHIV in each of these purposefully diverse populations.

XPRES study design and participants for prediction tool development

XPRES was a stepped wedge CRT with a retrospective baseline component conducted at 22 health facilities, including 5 hospitals and 17 clinics, which were purposively selected to be

representative of HIV treatment clinics in Botswana [15]. In the prospective, stepped wedge portion of the trial, all nonincarcerated, consenting, HIV–positive persons not yet on ART, regardless of TB treatment or symptom status, presenting to the study clinics between August 2012 and end of March 2014, were eligible for enrollment. This analysis was limited to adolescents and adults (aged \geq 12 years old), prospectively enrolled in the XPRES trial without a known TB diagnosis upon arrival at the study clinics (S1 Fig).

XPRES procedures

Per Botswana national guidelines during the time period of the study, all XPRES study participants were eligible for ART initiation if they had a CD4 count \leq 350 cells/µL, were diagnosed as having WHO stage III/IV, or were pregnant or breastfeeding [23]. All study participants received clinical care and follow-up appointments per Ministry of Health and Wellness guidelines, which included WHO TB symptom screening at the first and all subsequent clinic visits (S1 Table).

Interventions. The prospective XPRES cohort was recruited within 2 phases of the stepped wedge trial. In the first phase, all prospective XPRES participants received 2 enhanced care interventions in addition to standard of care (SOC): (1) additional support for intensified TB case finding; and (2) intensified tracing for patients missing clinic appointments. In the second phase, the Xpert MTB/RIF assay was initiated in place of sputum smear microscopy for TB diagnosis. We have previously shown that there was no significant difference in TB case finding between the first and second prospective phases of XPRES (5% versus 6%), although the prevalence of microbiologically confirmed TB was higher in the post-Xpert study phase (51% versus 65%) [16]. Details of the enrollment, intensified TB case finding and retention interventions, and follow-up procedures are described in a Supporting information appendix (S1 Text). XPRES participants were followed for 12 months, or until the end of TB treatment, whichever was later. The final follow-up visits for XPRES enrollees were in June 2015.

Development and internal validation of the prediction model

A clinically useful prediction model should demonstrate accurate prediction of the outcome in data other than that in which the model was developed. Therefore, we split the XPRES dataset in an approximately 1:1 ratio into southern clinics (n = 11) and northern clinics (n = 11) with southern clinics serving as the derivation dataset and northern clinics as the model validation dataset. This nonrandom approach to splitting the dataset is preferable to randomly splitting patients into derivation and validation groups because it reduces the similarity of the 2 sets of patients and is a stricter test of the derived model [24].

Outcome. The outcome of interest for this analysis was new diagnosis of active TB (clinical or microbiologically confirmed), within 6 months of arrival at the HIV treatment clinic. In this manuscript, this outcome is referred to as undiagnosed prevalent TB at arrival at the HIV treatment clinic [18,25,26]. Active TB during this initial 6-month post-clinic enrollment period is considered prevalent rather than incident TB based on prior clinical cohort data showing that 87% of active TB cases identified in months 0 to 6 after HIV clinic enrollment could have been diagnosed at the HIV clinic enrollment visit [26]. In addition, data from Zimbabwe show that the mean duration of smear positivity prior to TB diagnosis among HIV–positive adults to be 18 to 33 weeks [25]. Precedent for this approach and definition have been previously published [18]. We implemented intensive efforts to ascertain true active TB disease among participants with TB case finding procedures previously published and provided in a Supporting information appendix (S1 Text) [15,16]. **Candidate predictor variables.** We selected candidate predictor variables for potential inclusion in the predictive models based on prior publications and the need for variables to be reproducible, objective, and readily available in resource-constrained clinic settings [27]. We considered variables known to be associated with active TB including age, sex, marital status, education level, employment status, previous/current work as a miner, smoking history, prior TB treatment, history of a TB contact in the last 24 months, presence and number of WHO TB symptoms, body mass index (BMI) (weight/height²), hemoglobin concentration, CD4 count, temperature at ART initiation in degrees Celsius, and respiratory rate at enrollment visit [18,28–30].

Logistic regression model approach. Within the derivation dataset, we performed univariable logistic regression analyses assessing the association of each variable with risk of prevalent active TB. Continuous variables were assessed for nonlinearity with log odds of TB using multivariable fractional polynomial (MFP) models, as well as by comparing Akaike information criteria (AIC) and Bayesian information criteria (BIC) between models with linear or fractional polynomial terms. Where nonlinearity was observed, the appropriate fractional polynomial terms were included in the logistic regression. We also examined scatter plots of untransformed and transformed continuous variables and risk of TB to assess inflection points that might inform appropriate categorization of continuous variables. Where inflection points were close to previously published cutoffs for categorizing continuous variables, the previously published approach was used.

For the multivariable logistic regression analysis, a complete case analysis was chosen because few data (<10%) were missing. To inform generation of a parsimonious multivariable model, we used a stepwise backward elimination approach, starting with all candidate variables and excluding variables sequentially if p > 0.01 using both automatic and manual approaches. Prior regression derived scores used p-value cutoffs of >0.05 [18,31,32]; however, there is no accepted standard p-value cutoff for backward or forward stepwise variable elimination approaches [27]. Because we aimed to generate a parsimonious model, to increase feasibility of the practical clinical score in LMIC clinic settings, we used a >0.01 cutoff in line with recommendations from Royston and colleagues [27,33]. We also explored if findings changed using a forward stepwise addition approach. Where 2 or more predictors were highly correlated, only 1 was selected, to simplify the prognostic model. Plausible interactions between covariates (e.g., between sex and BMI [34]) were assessed using the likelihood ratio test.

Random forest model approach. We first built a random forest model with all 15 possible candidate variables that were included in the backward stepwise elimination approach. We fit the model using the randomForest R package with 1,000 trees. We used the *bestmtry* function to identify the optimum number of variables to be randomly included in each of the 1,000 trees (R version 3.6.1, R Core Team, 2017, R Foundation for Statistical Computing, Vienna, Austria) [35]. We used this model to order the 15 variables according to importance in predicting TB as measured by the mean decrease in Gini value for each variable [36]. The Gini value is a measure of decision tree node purity [37]. To develop a single decision tree, the best split at each node is assessed by evaluating which cutoff gives the most homogenous classifications (i.e., lowest Gini impurity according to published formulas [37]). The mean decrease in impurity is the average of a variable's total decrease in node impurity, weighted by the proportion of samples reaching that node in each individual tree in the random forest. Therefore, high mean decrease in Gini value indicates higher variable importance (i.e., the variable was on average important in splitting nodes into groups that had TB versus did not have TB).

We compared results with the logistic regression to assess if potentially important discriminatory variables had been eliminated in the backward stepwise regression. To assess any potentially important loss of discrimination through eliminating variables to create a parsimonious model, and to assess potential differences in discriminatory capacity between model approaches, we compared area under the receiver operating characteristic (AUROC) curve values between modeling approaches (logistic regression versus random forest) and between variable selection approaches (models with all 15 variables selected versus parsimonious models). Information from the backward stepwise regression and random forest modeling was used to generate the final parsimonious model.

Internal validation of parsimonious model. In both the derivation and validation datasets, we assessed multivariable logistic regression model calibration graphically in a calibration plot [24] and statistically using the Hosmer–Lemeshow test. We also assessed discrimination using the AUROC values. AUROC values of 0.7 to 0.79, 0.8 to 0.89, and >0.9 are, respectively, considered acceptable, excellent, and outstanding discrimination [38].

Clinical score generation. The final multivariable model was used to generate a practical clinical score. For these models, continuous variables were categorized in a clinically meaningful manner based on their functional form and information from the published literature. Per published precedent, each beta coefficient from this logistic regression model was then rescaled to generate a clinical score by dividing each coefficient by the smallest positive model coefficient and rounding to the nearest integer [18,39]. The total number of points was summed for each participant to calculate their total clinical score. In a sensitivity analysis, we generated fully standardized coefficients using the *listcoef* command in STATA (Stata, 2009, Stata Statistical Software, Release 16, College Station, Texas, United States). We rescaled the standardized coefficients and assessed whether risk score discrimination varied between the scores derived from original beta coefficients versus fully standardized coefficients in XPRES, XPHACTOR, TBFT, and Gugulethu cohorts [40].

External validation of risk scores

To externally validate the clinical risk score, we used data collected independently from the XPHACTOR cohort [18], TBFT trial [19], and Gugulethu cohort [22]. Characteristics of these studies as relate to setting, clinic types, eligibility criteria, dates of enrollment, ART eligibility criteria, TB symptoms screening, TB case finding approaches, and definition of active prevalent TB for this analysis are described in a Supporting information appendix (S2 Table).

For both the XPRES cohort (combined derivation and validation datasets) and the 3 validation datasets, we explored how sensitivity, specificity, PPV, NPV, percentage screened into diagnostic test algorithms, and AUROC curve values varied with increasing clinical score in terms of predicting active prevalent TB and compared this screening accuracy and discrimination performance with the current WHO TB symptom screening rule. Three risk groups were created to visualize increasing active prevalent TB risk with increasing clinical score and the percentage of ART enrollees falling into each risk group. The NNS to detect 1 TB case was compared between WHO TB symptom screening rules and a range of clinical score cutoffs.

All logistic regression and clinical score validation analyses were conducted using STATA 16 (Stata, 2009, Stata Statistical Software, Release 16). All random forest plot analyses and analyses to assess the mean decrease in Gini value associated with candidate predictor variables were done with R version 3.6.1. (R Core Team, 2017, R Foundation for Statistical Computing). Evaluating accuracy of different approaches to screening for undiagnosed active TB among PLHIV and evaluating prevalence and predictors of TB were part of the approved protocol (S1 Protocol). Augmenting the logistic regression approach with use of a machine learning approach and risk score generation were not prespecified but were used to build confidence in the analytic approach and translate the prespecified analysis into a screening tool potentially useful for clinicians. The study is reported in concordance with the Transparent Reporting of a

multivariable prediction model for Individual Prognosis or Diagnosis (TRIPOD) guidance for multivariable prediction models (<u>S1 Checklist</u>).

Ethical review

Ethical approval for each of the source studies was obtained from the relevant ethics committees in the country of data collection and from the trial sponsors. All participants provided informed written consent, or where the enrollee could not read or write, witnessed verbal informed consent. Ethical approvals for XPRES were obtained from the US Centers for Disease Control and Prevention (CDC) Institutional Review Board (IRB), the Health Research and Development Division of the Health Research and Development Committee (HRDC) in Botswana, and the University of Pennsylvania IRB No.4. XPHACTOR was approved by the ethics committees at the University of the Witwatersrand, University of Cape Town, and the London School of Hygiene & Tropical Medicine. TBFT was approved by the research ethics committees of the University of the Witwatersrand and the London School of Hygiene & Tropical Medicine and the South African Medicines Control Council. The Gugulethu prospective cohort study was approved by the research ethics committees of the University of Cape Town and the London School of Hygiene & Tropical Medicines Control Council. The Gugulethu prospective cohort study was approved by the research ethics committees of the University of Cape Town and the London School of Hygiene & Tropical Medicine.

Results

From the XPRES cohort, 5,418 eligible adult (\geq 12 years old) study enrollees with complete data for candidate predictors were included in the analysis (Fig 1). Overall, 318 (6%) of 5,418 enrollees had undiagnosed prevalent active TB at HIV clinic registration and study enrollment. From this XPRES cohort, the internal derivation (N = 2,771) and validation (N = 2,647) datasets were created (Table 1). Key characteristics including median age (34 years), percentage female (67% to 68%), median CD4 (240 to 249 cells/µL), and prevalence of active TB at enrollment (5% to 7%) were similar between XPRES derivation and validation datasets (Table 1).

Variable importance in logistic regression and random forest models

Table 2 summarizes the results of univariable and multivariable logistic regression model development. Although age (linear continuous variable), education level achieved, prior/ current work as a miner, previous TB treatment, and respiratory rate (transformed variable) were associated with prevalent TB in univariable analysis, these variables were eliminated in the stepwise backward elimination approach due to *p*-values in multivariable analysis >0.01. The full 15-variable multivariable logistic regression model is in a Supporting information appendix (S3 Table). Both backward stepwise elimination and forward stepwise addition approaches yielded the same results in terms of final variables selected, adjusted odds ratios, and *p*-values.

Rankings of variable importance as measured by size of beta coefficient in the logistic regression model (S4 Table) and by mean decrease in Gini value in the random forest model (S2 Fig) were similar with respect to presence or absence of WHO TB symptoms, temperature, and BMI, with these variables among the most important predictors. Notably, the transformed term of BMI, which was eliminated in the backwards stepwise logistic regression at p = 0.408, was considered the second most important variable according to the mean decrease in Gini value approach (52.766) and third most important in terms of size of the logistic regression beta coefficient (1.617). Given the importance of BMI in the variable ranking approach, importance in the published literature, and availability in resource-constrained clinics, BMI was retained in final multivariable model (Table 2). The final multivariable model included sex,



Fig 1. Study profiles for the derivation and validation datasets. ART, antiretroviral therapy; PLHIV, people living with HIV; SA, South Africa; TB, tuberculosis; TBFT, TB Fast Track.

https://doi.org/10.1371/journal.pmed.1003739.g001

smoking history, presence/absence of ≥ 1 WHO TB symptom, BMI (as a transformed term per the MFP analysis), temperature (modeled as 2 transformed terms per the MFP analysis), and hemoglobin concentration (continuous, linear term) (Table 2). There was no substantial correlation between measured temperature in its transformed or linear form and reported fever (Pearson correlation coefficient (r) = 0.1316) or between measured temperature and ≥ 1 reported WHO TB symptom (r = 0.0846).

Table 1. Comparison of derivation and validation datasets (internal and external).*

			Derivati souther	on dataset (Botswana rn clinics: <i>N</i> = 2,771)	Validation dataset (Botswana northern clinics: <i>N</i> = 2,647)		External validation dataset (SA, XPHACTOR: <i>N</i> = 1,807)		External validation dataset (TBFT, SA: <i>N</i> = 793)		External validation dataset (Gugulethu cohort, CT, SA: N = 488)	
Demo	graphics		n	(Median or %)	n	(Median or %)	n	(Median or %)	n	(Median or %)	n	(Median or %)
	Age (years) n, med	lian (IQR)	2,771	34.3 (28.8 to 41.3)	2,647	33.5 (28.3 to 40.8)	1,807	40.0 (34.0 to 47.0)	793	38.0 (32.0 to 45.0)	488	33.6 (27.9 to 40.7)
	Female, n, %		1,862	67%	1,790	68%	1,290	71%	424	53%	310	64%
	If female, pregnan	it, n, %	499	27%	568	32%	0	0%	0	0%	0	0%
	Marital status, <i>n</i> ,	Married/ civil union	306	11%	242	9%						
		Single	2,353	85%	2,322	88%						
		Widowed /divorced	112	4%	83	3%						
	Smoking history (ever), n, %	466	17%	575	22%	388	21%	180	23%	185	38%
	Employed, n, %		1,467	53%	1,023	39%						
	Education, n, %	None	154	6%	235	9%						
		Primary	637	23%	614	23%						
		Secondary	1,689	61%	1,597	60%						
		Higher	291	11%	201	8%						
	Ever a miner,		124	4%	143	5%						
HIV/	TB history											
	Taking ART at stu enrollment	ıdy	0	0%	0	0%	1,612	89%	0	0%	0	0%
	Previous TB treats	ment, <i>n</i> , %	232	8%	169	6%					130	27%
	TB contact in last <i>n</i> , %	24 months,	266	10%	230	9%						
	WHO TB sympton	ms, n, %	1									
		Cough	533	19%	466	18%	364	20%	424	53%	243	50%
		Weight loss	533	19%	577	22%	243	13%	621	78%	331	68%
		Fever	262	9%	223	8%	105	6%	269	34%	139	28%
		Night sweats	257	9%	243	9%	135	7%	297	37%	199	41%
	Number of V TB symptoms,	VHO 0 n, %	1,979	71%	1,837	69%	1,299	72%	133	17%	67	14%
		1	349	13%	397	15%	384	21%	173	22%	127	26%
		2	200	7%	203	8%	136	8%	191	24%	135	28%
		3	136	5%	134	5%	41	2%	145	18%	121	25%
		4	107	4%	76	3%	17	1%	151	19%	38	8%
	Duration of WHO symptoms	n, median (IQR)	792	60 (30 to 120)	810	60 (21 to 150)						
Clinic	al characteristics											
	CD4 ⁺ T-cell count (cells/µL)	n, median (IQR)	2,771	240 (131 to 314)	2,647	249 (151 to 321)	1,807	400 (246 to 600)	793	73 (34 to 109)	488	167 (95 to 231)
	Weight (kg)**	n, median (IQR)	2,771	59 (52 to 69)	2,647	60 (53 to 69)			793	57 (50 to 66)	488	64 (56 to 73)
	BMI (kg/m ²)	n, median (IQR)	2,771	21.8 (19.2 to 25.4)	2,647	21.5 (18.9 to 24.7)	1,807	25.0 (21.4 to 29.3)	793	20.9 (18.6 to 24.5)	488	23.5 (20.9 to 27.1)
	Hemoglobin g/dL	n, median (IQR)	2,771	11.9 (10.5 to 13.3)	2,647	12.0 (10.7 to 13.4)	1,807	13.1 (11.8 to 14.3)	793	11.1 (9.6 to 12.8)	488	12.0 (10.6 to 13.4)
	Temperature (°C)	n, median (IQR)	2,771	36.2 (35.8 to 36.7)	2,647	36.1 (35.7 to 36.5)			793	36.3 (36.0 to 36.6)		
Respiratory rate <i>n</i> , median (breaths/min) (IQR)		2,771	20 (18 to 21)	2,647	18 (17 to 20)							
New TB diagnosis												
	Cumulative pro	evalent active TB, <i>n</i> , %	189	6.8%	129	4.9%	83	4.6%	77	9.7%	90	18.4%
mici	Cumulat obiologically confirm	ive incidence med TB, <i>n</i> , %	96	3.5%	64	2.4%	62	3.4%	77	9.7%	90	18.4%
Time to diagnosis of <i>n</i> , median prevalent TB (days) (IQR)		n, median (IQR)	189	16 (7 to 35)	129	19 (4 to 48)						

* Where variable is blank, the data were not collected or not provided from the source study for this analysis.

** BMI was used as the covariate for nutritional status rather than weight (weight was not considered as an independent predictor).

BMI, body mass index; CT, Cape Town; IQR, interquartile range; SA, South Africa; TB, tuberculosis; TBFT, TB Fast Track; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1003739.t001

			Not diagnosed with TB (N = 2,582)		Di	Diagnosed with TB		Unadjusted		Final adjusted regression		
					within 6 months (N = 189)		OR	95% CI	<i>p</i> -value	AOR	95% CI	<i>p</i> -value
Demographics		n	Median(IQR)/%	n	Median(IQR)/%							
	Age, years (for every 10-year incre	ase)		34 (29 to 41)		38 (32 to 44)	1.24	(1.15 to 1.33)	< 0.001			
	Sex	Female	1,768	95%	94	5%	1.00			1.00		
		Male	814	90%	95	10%	2.20	(1.63 to 2.95)	< 0.001	1.91	(1.27 to 2.88)	0.002
	Marital status	Married/civil union	288	94%	18	6%	1.00					
		Single	2,190	93%	163	7%	1.19	(0.71 to 2)	0.508			
		Widowed/divorced	104	93%	8	7%	1.23	(0.44 to 3.41)	0.690			
	Smoking history (ever smoked)	No	2,165	93%	154	7%	1.00			1.00		
		Yes-ever smoked	417	89%	50	11%	1.82	(1.56 to 2.12)	< 0.001	1.44	(1.12 to 1.85)	0.004
	Employed	Employed	1,370	93%	97	7%	1.00					
		Unemployed	1,212	93%	92	7%	1.07	(0.71 to 1.62)	0.742			
	Education	None	137	89%	17	11%	1.00					
_		Primary	588	92%	49	8%	0.67	(0.38 to 1.18)	0.166			
		Secondary	1,576	93%	113	7%	0.58	(0.35 to 0.95)	0.030			
		Higher	281	97%	10	3%	0.29	(0.14 to 0.58)	< 0.001			
	Ever a miner	No	2,478	94%	169	6%	1.00					
		Yes	104	84%	20	16%	2.82	(2.04 to 3.9)	< 0.001			
Н	IIV/TB history											
	Previous TB treatment	No	2,381	94%	158	6%	1.00					
		Yes	201	87%	31	13%	2.32	(1.43 to 3.78)	0.001			
	Any TB contact in last 24 months	No	2,339	93%	180	7%	1.00					
		Yes	243	91%	24	9%	1.33	(0.83 to 2.14)	0.233			
_	Number of WHO symptoms	0	1,936	98%	43	2%	1.00					
_		> = 1	646	82%	146	18%	10.18	(6.79 to 15.26)	< 0.001	6.91	(4.55 to 10.49)	< 0.001
Clinical characteristics												
	CD4 (per 10-cell increase)		2,582	247 (139 to 316)	189	151 (57 to 255)	0.96	(0.94 to 0.97)	< 0.001			
_	Weight (per 1-kg increase) ^a		2,582	59.4 (52.3 to 69.2)	189	53.7 (47.0 to 62.0)	0.97	(0.95 to 0.99)	0.001			
_	BMI (per 1-unit increase) ^b		2,582	21.9 (19.4 to 25.5)	189	19.4 (17.2 to 22.3)	0.90	(0.83 to 0.97)	0.004	0.98	(0.93 to 1.05)	0.612
Hemoglobin (per 1g/dL increase)		2,582	12.0 (10.6 to 13.3)	189	10.6 (9.2 to 12.3)	0.76	(0.69 to 0.83)	< 0.001	0.78	(0.7 to 0.86)	< 0.001	
Temperature at enrollment (per 1°C increase) ^c			2,582	36.2 (35.8 to 36.6)	189	36.4 (36.0 to 37.1)	2.13	(1.57 to 2.88)	< 0.001	1.46	(1.18 to 1.81)	< 0.001
RR (breaths/min) ^d		2,582	20 (18 to 20)	189	20 (18 to 22)	1.03	(1.01 to 1.05)	0.010				

Table 2. Univariable and multivariable logistic regression analysis in the derivation dataset (N = 2,771).

^a Due to correlation between weight and BMI (r = 0.8837), weight was not included in the stepwise backward regression, because BMI is a better measure of nutritional status than weight alone.

^b Due to nonlinearity in the association between BMI and log odds TB, BMI was modeled as a transformed term from the MFP analysis (transformed BMI = $X^{-.5}$ -.666749355, where X = BMI/10). Output shown is for the single linear term to facilitate interpretation of average BMI effect (i.e., higher BMI associated with lower TB risk). In the backward stepwise regression, the *p*-value associated with BMI term was 0.4077 at point of elimination. Given the importance of BMI as a predictor in the random forest model (second most important predictor), ease of availability of this variable in almost all resource-limited clinics, and importance of BMI in published literature, BMI was retained in the final adjusted model.

^c Due to nonlinearity in the association between temperature and log odds TB, temperature was modeled as 2 transformed terms (term 1 = temperature -36.12674419; term 2 = temperat^2-1305.141645). Output shown is for the single linear term to facilitate interpretation of average temperature effect (i.e., higher temperature associated with higher TB risk). In the backward stepwise regression, the *p*-value associated with each transformed term was 0.005 and 0.004, respectively.

^d Due to nonlinearity in the association between RR and log odds TB, RR was modeled as a transformed term from the MFP analysis (transformed term = X^{-1}

-5.170636738, where X = RR/100). Output shown is for the single linear term to facilitate interpretation of average RR effect (i.e., higher RR associated with higher TB risk). In the backward stepwise regression, the *p*-value associated with the transformed term was 0.0251 at the point of elimination from the model.

AOR, adjusted odds ratio; BMI, body mass index; CI, confidence interval; IQR, interquartile range; MFP, multivariable fractional polynomial; OR, odds ratio; RR, respiratory rate; TB, tuberculosis; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1003739.t002

Internal validation of final multivariable regression model

For the derivation dataset, the Hosmer–Lemeshow statistic for the TB prediction model (p = 0.135) (S5 Table), and the calibration curve (Fig 2), indicated good model fit. Although the Hosmer–Lemeshow statistic for the internal validation dataset (p = 0.0001) indicated lack of fit with overestimation of prevalent TB, with 169 cases of prevalent TB predicted versus 129 observed, (1) the Hosmer–Lemeshow test is sensitive to sample size, and our sample size is large; and (2) the calibration curve (Fig 2) indicated adequate prediction performance for the 10 risk groups. In addition, the AUROC curve values for the derivation (0.839; 95% CI, 0.811 to 0.868) and validation datasets (0.799; 95% CI, 0.757 to 0.841) indicated excellent and borderline excellent discrimination, respectively (Fig 2).

Comparison of regression and random forest discrimination

Comparison of discriminatory performance between 15-covariate and 6-covariate parsimonious models (S3 Fig) indicated very little loss of discrimination by eliminating 9 of the covariates from the predictive model, building confidence in the final multivariate model. Similarly, although the random forest approach had far superior discrimination on the derivation dataset



Fig 2. Logistic regression model AUROC curves and calibration curves for the internal derivation and validation datasets, respectively. AUROC, area under the receiver operating characteristic; TB, tuberculosis.

https://doi.org/10.1371/journal.pmed.1003739.g002

versus the logistic regression approach, both modeling approaches had similar discrimination in the internal validation dataset (S3 Fig).

Transformation from regression model to clinical score

We used WHO-recommended cutoffs for severe anemia in adults (<8.0 g/dL) [41] and for being underweight (BMI<18.5 kg/m²) to categorize hemoglobin and BMI variables, respectively. Temperature was classified as \leq 37.5°C versus >37.5°C based on the observed distribution of TB prevalence risk as measured temperature increased and a common definition of a low-grade fever or higher (>37.5°C) (S4 Fig) [42]. The multivariable model with categorization of these continuous variables in the derivation dataset is presented in Table 3.

The final model, categorized in this way, retained excellent discrimination in the derivation dataset (AUROC 0.823; 95% CI, 0.793 to 0.853) and acceptable discrimination in the validation dataset (AUROC 0.771; 95% CI, 0.727 to 0.815), and the Hosmer–Lemeshow statistic *p*-values were 0.1940 in the derivation and 0.0002 in the validation datasets indicating similar goodness of fit as was observed prior to variable categorization. The clinical scores that could be used in clinic settings to identify those at risk of prevalent active TB are illustrated in Fig 3.

External validation of risk scores

The clinical score for each predictor was generated and applied to each external dataset, where the possible range for the total score was 0 to 20 (S6 Table). Performance of the clinical score at different cutoffs, in terms of sensitivity, specificity, NPV, PPV, and the percentage of clinic enrollees that would be offered a TB test is provided in a Supporting information appendix (S5 Fig). Across the 4 datasets, a clinical score of \geq 7 would give similar sensitivity and specificity to WHO 4-symptom TB screening rule. Moving the clinical score to \geq 2 would give superior sensitivity versus WHO 4-symptom TB screening rule, but with some loss of specificity. For example, sensitivity in detecting prevalent active TB using WHO 4-symptom TB screening rule was 73%, 80%, 94%, and 94% in XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively, but this increased to 88%, 87%, 97%, and 97%, when a clinical score of \geq 2 was used. However, specificity would decline from 73%, 70%, 18%, and 16% if WHO 4-symptom TB-screen was used to 55%, 58%, 13%, and 12% if the clinical score of \geq 2 was used. Similarly,

Table 3. Multivariable model and clinical score in the derivation dataset (N = 2,771).

Predictor		AOR	95% CI	<i>p</i> -value	β coefficient	Score
WHO TB symptoms	No symptoms	1.00				
	\geq 1 symptom	7.00	(4.66 to 10.52)	<0.001	1.95	7
Sex	Female	1.00				
	Male	1.35	(0.88 to 2.08)	0.173	0.30	1
Smoker	Never	1.00				
	Ever smoked	1.32	(1.03 to 1.7)	0.030	0.28	1
Hemoglobin	$\geq 8 \text{ g/dL}$	1.00				
	<8 g/dL	2.50	(1.28 to 4.85)	0.007	0.91	3
Temperature	≤37.5	1.00				
	>37.5	5.53	(3.5 to 8.72)	<0.001	1.71	6
BMI	≥18.5	1.00				
	<18.5	1.70	(1.12 to 2.59)	0.013	0.53	2

AOR, adjusted odds ratio; BMI, body mass index; CI, confidence interval; TB, tuberculosis; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1003739.t003

Risk Factor	Category	Associated points	Assigned score
	Female	0	
Sex	Male	1	
No. of WHO TB	Zero	0	
(cough, fever, weight loss, night sweats)	≥1	7	
Smaller	Never	0	
Smoker	Ever smoked	1	
T (°C)	≤37.5	0	
Temperature (°C)	>37.5	6	
DA1(1) = (-2)	≥18.5	0	
BIVII (Kg/m²)	<18.5	2	
Hemoglobin Level	≥8.0	0	
(g/dL)	<8.0	3	_
		Total	
	+		
<2 = low risk	2–10 = moder	ate risk	>10 = high risk

Fig 3. Clinical score for predicting TB among PLHIV. BMI, body mass index; PLHIV, people living with HIV; TB, tuberculosis; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1003739.g003

the percentage of patients screened into a TB diagnostic test algorithm (referred to as "screen in" in S5 Fig) per WHO 4-symptom TB screening rule would be 30%, 32%, 83%, and 86% in the XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively, but this increases to 45%, 42%, 87%, and 88% if a clinical score of ≥ 2 is used.

Notably, when the XPHACTOR dataset was restricted to clients on ART for >3 months, the clinical score retained good sensitivity and moderate specificity (S6 Fig). For example, at clinical score \geq 2, sensitivity was 80% and specificity 60% versus WHO 4-symptom screening criteria that provided 69% sensitivity and 72% specificity.

The NPV of WHO 4-symptom TB screen was 97.7% and 98.6%, 96.2%, and 92.5% in the XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, increasing to 98.7%, 98.9%, 97.8%, and 94.2% when the clinical score at cutoff \geq 2 was used, reflecting a 1%, 0.3%, 1.6%, and 1.7% increase in NPV.

When restricting the XPRES and TBFT cohorts to those who died within 6 months of clinic enrollment, the clinical score at a cutoff of \geq 2 had superior sensitivity to WHO 4-symptom TB screen in predicting TB in the XPRES cohort (94% versus 79%) and similar sensitivity in the TBFT cohort (100% versus 100%) (S7 Fig). However, specificity of the clinical score at \geq 2 was inferior to that of WHO 4-symptom TB screen in both XPRES (16% versus 31%) and TBFT (3% versus 8%) cohorts.

Overall, the clinical score had superior discrimination in the XPRES and XPHACTOR datasets than in the TBFT and Gugulethu cohorts (S8 Fig). The XPRES and XPHACTOR cohorts were more similar with respect to median baseline CD4 count (245/ μ L in XPRES and 400/ μ L in XPHACTOR) compared with TBFT and Gugulethu cohorts (73/ μ L in TBFT and 167/ μ L in Gugulethu cohorts) (Table 1). Similarly, XPRES and XPHACTOR cohorts were more similar with respect to baseline prevalence of active prevalent TB (6% in XPRES and 5% in XPHACTOR) compared with TBFT and Gugulethu cohorts (10% in TBFT and 18% in Gugulethu cohort).

In 4 sensitivity analyses, we compared discrimination of our final risk score, derived from the multivariable model beta coefficients described in Table 3 versus (1) risk scores derived from fully standardized beta coefficients; (2) risk scores excluding BMI (the variable identified as important by random forest model variable ranking); (3) risk scores excluding hemoglobin (the only variable requiring a blood test); and (4) risk scores excluding sex and smoking (variables assigned only 1 point in the risk score). In all sensitivity analyses, our final risk score had modest but consistently superior discrimination across XPRES, XPHACTOR, TBFT, and Gugulethu cohorts (S7 Table).

Risk scores were grouped into low (<2), moderate (2 to 10), and high-risk categories (>10) (Fig 4). Prevalence of active TB among enrollees in low-, moderate-, and high risk groups was 1%, 3%, and 33% among XPRES enrollees, 1%, 11%, and 22% among XPHACTOR enrollees, 2%, 8%, and 26% for TBFT enrollees, and 6%, 19%, and 32% for Gugulethu cohorts, respectively, indicating a differentiation of prevalent TB risk by the respective clinical scores.

NNS to diagnose one TB case

In the cohorts with the highest prevalence of active TB (TBFT and Gugulethu), a clinical score with cutoff of ≥ 2 would give a marginally higher NNS to diagnose one TB case compared with the 4-symptom WHO screen (Fig 5); the NNS increased from 8.6 to 9.3 in TBFT and from 4.7 to 5.0 in Gugulethu cohorts. In contrast, in cohorts with lower prevalence of active prevalent TB (XPPRES and XPHACTOR), the NNS increased to a larger extent (from 5.0 to 9.3 in XPRES and from 6.9 to 11.0 in XPHACTOR). The NNS in the highest risk group with clinical score >10 (i.e., equivalent to ≥ 11 in Fig 5) was uniformly low being 3.0, 4.5, 3.9, and 3.1 in XPRES, XPHACTOR, TBFT, and Gugulethu cohorts, respectively. If the NNS threshold was set at about 5.0, this would correspond to clinical scores of about ≥ 9 to 10 in XPRES, XPHACTOR, and TBFT cohorts, but ≥ 2 in the Gugulethu cohort.

Discussion

To our knowledge, this study provides new information by deriving and externally validating an initial clinical score for active TB among both ART-naive and ART-experienced adult PLHIV that includes but does not rely solely on WHO TB symptom screening and allows flexibility in choosing the desired sensitivity, specificity, NPV, PPV, and NNS across a range of cutoffs, depending on the setting, use case scenario, and population served. In addition, following further validation and evaluation steps, the screening tool could potentially be used to reduce the likelihood of missing subclinical TB, which accounted for 6% to 27% of all TB cases across

35%

30%

25%

20%

15%

10%

5%

0%

22%

Risk of Active TB

XPRES Cohort Botswana (N=5,418)

XPHACTOR Cohort SA (N=1,811)

1,400

400

TB





200 1% <2 2 - 10 >10 Active Prevalent TB □No Active Prevalent TB -Risk of Active Prevalent TB







https://doi.org/10.1371/journal.pmed.1003739.g004

studies; this could potentially help reduce morbidity and mortality due to late or missed TB diagnosis and reduce TPT prescription to PLHIV needing a full TB treatment course. Similarly, following further validation efforts, the screening tool's differentiation of 3 risk groups could be used to inform differentiated care in LMIC clinic settings, which could potentially improve efficiency and impact morbidity and mortality. Finally, the different modeling



WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1003739.g005

approaches provide unique insight into covariate predictor importance and practical ways machine learning can be helpful in predicting TB.

While 5 previous studies have generated clinical scores for TB among PLHIV, 3 were designed as a second step after screening positive using WHO 4-symptom TB screening rule [18,32,43], and 2 generated a relatively complex score (13 signs and symptoms), were focused

on ART-naive patients only in Bissau, and lack external validation [44,45]. One recent TB clinical score was derived from a cohort of HIV-positive and HIV-negative patients in SA who had symptoms of active TB; however, this score would not be able to detect asymptomatic active TB among PLHIV [46]. Our tool was deliberately designed to use widely available variables in LMIC settings and was externally validated using 3 different cohorts. The only blood test needed for our score is hemoglobin concentration. Point-of-care (POC) hemoglobin measurement devices are widely available, durable, easy to use, have good accuracy [47,48], are useful for non-HIV-related care, and are inexpensive [49]. Noninvasive transcutaneous spectrophotometry solutions for hemoglobin measurement are also available but require further evaluation of accuracy, feasibility, and acceptability [50-52]. In addition, the importance of severe anemia as a predictor of active TB [53], as well as the biological mechanism (i.e., hepcidin-driven iron sequestration in the reticuloendothelial system), has been well described [54,55]. Other studies have examined the screening accuracy of C-reactive protein (CRP) compared with WHO 4-symptom TB screen with results differing between studies, settings, and population groups [56-58]. A recent meta-analysis comparing CRP to WHO 4-symptom screen showed comparable sensitivity and higher or similar specificity [59]. The key barrier to use of CRP is that access to POC CRP blood tests is currently very limited [57]. As access to CRP tests expand, future evaluations including CRP in similar risk scores is needed [60].

Measured temperature at $>37.5^{\circ}$ C, an additional variable routinely available in LMIC clinic settings, was also independently predictive of TB, indicating the importance of objective measures of fever in addition to patient history [61]. The lack of correlation between reported fever and measured temperature might be due to lack of subjective experience of fever, especially if the fever is chronic or low grade, incomplete understanding of the symptom screening question by the client living with HIV, or patient preference not to acknowledge subjective feelings of fever [44,45].

While the use of electronic medical records (EMRs) are expanding in resource-constrained settings, POC EMR systems that can run regression models retaining continuous rather than categorized variables [62] are not widely accessible in LMIC settings due to logistical and resource constraints of equipment procurements and maintenance, limited and inconsistent access to electricity, and computer literacy capacity [63,64]. Our score that could be used in both paper-based approaches and simple EMR solutions is therefore appropriate for the setting it is designed for [18,39].

A potential advantage of our clinical score over WHO 4-symptom screening tool is that the score could be used by program managers to choose the desired cutoff with associated sensitivity, specificity, NPV, PPV, and NNS. For example, among people starting or restarting ART, among whom mortality risk from undiagnosed disseminated TB remains relatively high, a more sensitive screening tool could help reduce morbidity and mortality [2,65]. Notably, in the XPRES cohort, sensitivity of detecting TB at the initial HIV clinic visit among those who died within 6 months of clinic enrollment increased from 79% with WHO 4-symptom rule to 94% with our clinical score at cutoff ≥ 2 , suggesting the potential for improved early case finding with possible morbidity and mortality reductions [14,16]. However, further evaluation and validation of the screening tool is needed.

In addition, with support from global health donors, many countries are embarking on ambitious TPT scale-up for PLHIV, with the majority of targeted TPT recipients being longterm stable ART patients [9]. Following the 2018 United Nations high-level meeting on TB, the US President's Emergency Plan for AIDS Relief (PEPFAR) committed to reaching >13 million PLHIV with TPT by 2021 [9]. Although increases in NPV by using a clinical score cutoff of \geq 2 instead of WHO 4-symptom TB screen are modest, ranging from 0.3% to 1.7%, use of the clinical score cutoff of \geq 2 during the proposed TPT scale-up for PLHIV could potentially avoid 19,500 to 97,500 PLHIV with active TB (assuming 5% TB prevalence) or 39,000 to 195,000 PLHIV (assuming 10% TB prevalence) being inappropriately prescribed TPT. Missed active TB increases morbidity and mortality risk for the patient, but also increases risk of isoniazid-resistant TB [9], which is associated with worse treatment outcomes and may be transmitted to others [10,66]. Our simple screening rule approach to increasing sensitivity and NPV would be much less expensive and logistically challenging than the current WHO recommendation to consider adding CXR, which is not widely available at LMIC clinic settings, to WHO symptom screen [11,67]. In addition, the NPV increase associated with adding CXR to WHO screening rule of 0.9% (at 5.0% TB prevalence) is similar to the NPV increase gained by our much simpler and less costly clinical score at cutoff ≥ 2 (0.3% to 1.7% at TB prevalence of 4.6% to 18.4%) [6].

Another potential advantage of the clinical score is that the cutoff can be tailored to the use case scenario [68]. As described above, for clients at ART enrollment, reenrollment, or being assessed for TPT eligibility, ruling out active TB is a high priority, and, therefore, high sensitivity and NPV are desired and a cutoff of \geq 2 could be chosen. However, for stable patients on long-term ART who have completed a course of TPT, a screening rule cutoff with higher specificity and therefore higher PPV (e.g., cutoff >10) could be chosen to lower the NNS and improve efficiency and cost-effectiveness [18,69,70].

The clinical score could also facilitate differentiated TB care based on TB risk [71]. Firstly, the clinical score is relatively simple and could be used by community healthcare workers in the community [71], with community-based care models for HIV and TB increasingly important to decongest health facilities during the Coronavirus Disease 2019 (COVID-19) pandemic [72]. Secondly, the score could facilitate identifying which new or long-term ART patients should be prioritized for dedicated adherence and retention resources to ensure completion of the TB diagnostic and treatment cascade, with loss to follow-up from HIV–TB care a common problem in LMIC [16,73,74]. Similarly, prioritization of limited on-site GeneXpert diagnostics can be informed by the clinical score to increase cost-effectiveness of POC Xpert use [75]. Finally, diagnostic and therapeutic algorithms could be stratfied by risk groups, with more aggressive TB case finding and treatment approaches appropriate for highest risk groups (e.g., sputum culture, urinary diagnostics, abdominal sonography, or empiric TB treatment) [19]. For example, for patients who are severely ill, and where TB diagnostics are not available or have long turnaround times, a high clinical score of >10, which has a PPV of 22% to 33%, could guide rapid initiation of empiric TB treatment [76].

The dual modeling approach of logistic regression and use of random forest machine learning helps to build confidence in the final practical clinical score for use in LMIC clinic settings. The random forest approach, similar to other machine learning approaches, is better able to capture nonlinear relationships between predictors and outcomes compared with well-established generalized linear regression models [17], because random forest models are not dependent on making assumptions of average linear or curvilinear associations between covariates and outcomes. Among machine learning models, random forest models are particularly strong at predicting categorical outcomes like our TB outcome [77]. For example, despite using the fractional polynomial transformed BMI variable in the logistic regression backward stepwise elimination approach, it was eliminated from the parsimonious model at p > 0.01. In contrast, the importance of BMI in discriminating prevalent active TB using the mean decrease in Gini value analysis indicated the importance of BMI in its ability to accurately split groups of patients into those who have or do not have prevalent active TB across the 1,000 decision trees examined in our random forest model. The high ranking of BMI according to mean decrease in Gini value indicates the significant decrease in average, weighted decision tree node purity that occurred when BMI was removed from the possible list of predictor variables [77].

Our analysis also indicates a potential weakness of machine learning approaches. Although the random forest model is superior to single decision trees in reducing the likelihood of overfitting to the training data [77], we observed extremely high discrimination of the random forest model on the training data and a significant drop in discrimination on the validation data. This highlights the importance of a stringent validation approach [17]. Notably, the most widely available training resources and publications use a random 75%:25% split to create training and validation datasets for random forest models [36], but we purposefully split the dataset into northern and southern clinics in Botswana with a 50%:50% split, which is in line with more stringent validation approaches that help better assess generalizability of predictive models [17,24]. The tendency of the machine learning approach used in this study to overfit to the training data is therefore a limitation of the study. However, this joint approach of using multiple modeling approaches represents a useful contribution to the TB screening literature in line with emerging expert guidance [78,79].

Components of this study that build confidence in the results include the use of data from high-quality prospective cohorts, meaning there was minimal missing covariate data and strong ascertainment of the primary outcome of interest (prevalent active TB) through strengthened ICF in XPRES throughout follow-up and baseline collection of sputum samples for TB diagnosis in all 3 validation cohorts. The study observed high screening accuracy in the 3 external validation cohorts, XPHACTOR, TBFT, and Gugulethu cohorts, representing 3 geographically separate cohorts, with very different cohort characteristics, in very different settings. Limitations include that the risk score has been validated in adult cohorts of PLHIV in sub-Saharan Africa (SSA) and may not be generalizable to pediatric cohorts or cohorts in resource-rich settings like the US and Europe. In addition, our goal was to validate the simple clinical risk score in external validationd datasets, rather than the statistical models; therefore, only the clinical risk scores should be considered externally validated in this analysis. In addition, further evaluation and validation of this approach is needed in SSA, especially in East and West African cohorts. Since only 1 cohort (XPHACTOR) included patients currently on ART, further evaluation in cohorts of long-term ART enrollees would be helpful. Another limitation is that the approaches to TB case finding were different across the 4 cohorts and that for the XPRES and XPHACTOR cohorts a clinical definition of TB was included in the TB outcome definition, whereas for TBFT and Gugulethu cohorts, results of enrollment sputum collection for TB culture and Xpert were used to define the TB outcome. However, model results did not change substantially when we restricted the TB outcome in XPRES and XPHACTOR datasets to microbiologically confirmed TB [18]. In addition, our score does not replace WHO 4-symptom screening rule for active TB, but rather supplements WHO screening rule with additional data variables to help detect asymptomatic (subclinical) TB, allow for flexibility in choosing cutoffs, and allow differentiated care. Additional evaluation is needed to further determine feasibility and added benefit of collecting the additional variables.

Another limitation is that in XPRES, sputum samples for microbiological diagnosis were only obtained from symptomatic XPRES enrollees at enrollment and during follow-up. Although the intensive TB symptom and ICF cascades during repeat visits during 6 months of follow-up in XPRES make it less likely that active TB disease was missed over the course of 6 months of follow-up [26], persistently, subclinical TB disease would have been missed. In addition, in the XPHACTOR dataset, 45% of study participants were excluded from the validation dataset due to missing hemoglobin, and in the TBFT dataset, 29% were excluded because they could not produce sputum for culture at trial enrollment. Comparisons of available patient characteristics between persons excluded versus included in the validation datasets did not indicate notable differences, but additional validation exercises in contemporary cohorts with complete covariates are warranted to further build confidence in the risk score. In conclusion, following further validation and evaluation steps, this new, simple TB screening clinical score for PLHIV, which is appropriate for both ART-naive and ART-experienced PLHIV in SSA, and which incorporates but does not solely rely on WHO 4-symptom screening rule, is a potential timely addition to practical TB screening approaches in LMIC. The clinical score improves on WHO 4-symptom screening rule's capacity to detect subclinical TB, carrying potential associated morbidity, mortality, and TB transmission reduction benefits [5]. The clinical score provides improved sensitivity and NPV over WHO 4-symptom TB screen, which is needed ahead of intensive global TPT scale-up efforts. Finally, the range of clinical scores allows clinicians and program managers to differentiate patient care and choose cutoffs based on the use case scenario and availability of resources to improve precision and quality of patient-centered care.

Supporting information

S1 Fig. XPRES trial profile. (TIF)

S2 Fig. Random forest model variable importance ranking by mean decrease in Gini value in the derivation dataset (N = 2,771). (TIF)

S3 Fig. Comparison of AUROC curves by modeling approach (logistic regression versus random forest), covariate number (15 versus 6 variables), and in derivation versus validation datasets. AUROC, area under the receiver operating characteristic. (TIF)

S4 Fig. Association between temperature modeled as a transformed versus linear variable and prevalent active TB. TB, tuberculosis. (TIF)

S5 Fig. Distribution of clinical risk scores and associated sensitivity, specificity, NPV, and PPV for TB across 4 study cohorts. NPV, negative predictive value; TB, tuberculosis. (TIF)

S6 Fig. Sensitivity, specificity, NPV, and PPV of TB clinical score versus WHO 4-symptom TB screening rule among ART-experienced clinic attendees (XPHACTOR, *n* = 1612). ART, antiretroviral therapy; NPV, negative predictive value; PPV, positive predictive value; TB, tuberculosis; WHO, World Health Organization. (TIF)

S7 Fig. Sensitivity, specificity, NPV, and PPV of clinical score versus WHO 4-symptom screening rule among those who died within 6 months of clinic enrollment. NPV, negative predictive value; PPV, positive predictive value; WHO, World Health Organization. (TIF)

S8 Fig. Clinical score discrimination according to AUROC curve by study cohort. AUROC, area under the receiver operating characteristic. (TIF)

S1 Table. HIV care clinical follow-up of clients in the Botswana XPRES cohort (2010 to 2015).

(PDF)

S2 Table. Comparison of XPRES and external validation datasets. (PDF)

S3 Table. A 15-variable multivariable model in derivation dataset (*N* = 2,771). (PDF)

S4 Table. Importance of predictors in logistic regression versus random forest modeling approaches.

(PDF)

S5 Table. Hosmer–Lemeshow test for calibration of final TB prediction model. TB, tuberculosis.

(PDF)

S6 Table. Performance of the TB prediction clinical score in derivation and validation datasets. TB, tuberculosis. (PDF)

S7 Table Sensitivity analyses comparing discrimination between our final risk score and

risk scores that (a) were derived from fully standardized bet coefficients, (b) excluded BMI, (c) excluded hemoglobin, and (d) excluded sex and smoking variables BMI, body mass index.

(PDF)

S1 Text. XPRES enrollment and follow-up procedures. (PDF)

S1 Protocol. XPRES trial protocol. (PDF)

S1 Checklist. TRIPOD checklist: Prediction model development and validation. TRIPOD, Transparent Reporting of a multivariable prediction model for Individual Prognosis or Diagnosis. (PDF)

Acknowledgments

The authors would like to thank all study participants, participating health facilities, and project team members for their contributions to improving TB and HIV care in Botswana. We also thank Prof. Stephen D. Lawn for his contributions as a mentor, advisor, and subject matter expert prior to his death on September 23, 2016. We also thank Sherri L. Pals for her contributions to the statistical design of XPRES and statistical advice in developing analysis plans.

Disclaimers: The findings and conclusions in this report are those of the authors and do not necessarily represent the official position of the US Centers for Disease Control and Prevention.

Author Contributions

Conceptualization: Andrew F. Auld, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.

Data curation: Andrew F. Auld, Andrew D. Kerkhoff, Yasmeen Hanifa, Robin Wood, Salome Charalambous, Tefera Agizew, Christopher Serumola, Alison D. Grant, Katherine Fielding.

Formal analysis: Andrew F. Auld.

- **Funding acquisition:** Andrew F. Auld, Rosanna Boyd, Alyssa Finlay, James C. Shepherd, Tedd V. Ellerbrock, Alison D. Grant.
- **Investigation:** Andrew F. Auld, Tefera Agizew, Anikie Mathoma, Rosanna Boyd, Anand Date, Unami Mathebula-Modongo, Heather Alexander, Goabaone Rankgoane-Pono, Pontsho Pono, Alyssa Finlay, James C. Shepherd, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.
- **Methodology:** Andrew F. Auld, Yuliang Liu, Ray W. Shiraishi, Christopher Serumola, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.
- Project administration: Andrew F. Auld, Tefera Agizew, Anikie Mathoma, Rosanna Boyd, Anand Date, George Bicego, Unami Mathebula-Modongo, Heather Alexander, Christopher Serumola, Goabaone Rankgoane-Pono, Pontsho Pono, Alyssa Finlay, James C. Shepherd, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.
- **Resources:** Andrew F. Auld, Andrew D. Kerkhoff, Yasmeen Hanifa, Robin Wood, Salome Charalambous, Yuliang Liu, Tefera Agizew, Rosanna Boyd, Goabaone Rankgoane-Pono, Pontsho Pono, Alyssa Finlay, James C. Shepherd, Alison D. Grant, Katherine Fielding.
- **Software:** Andrew F. Auld, Yuliang Liu, Ray W. Shiraishi, Christopher Serumola, Alison D. Grant, Katherine Fielding.
- Supervision: Andrew F. Auld, Tefera Agizew, Anikie Mathoma, Rosanna Boyd, Anand Date, Ray W. Shiraishi, George Bicego, Unami Mathebula-Modongo, Heather Alexander, Goabaone Rankgoane-Pono, Pontsho Pono, Alyssa Finlay, James C. Shepherd, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.
- Validation: Andrew F. Auld, Andrew D. Kerkhoff, Yasmeen Hanifa, Robin Wood, Salome Charalambous, Yuliang Liu, Tefera Agizew, Anikie Mathoma, Anand Date, George Bicego, Unami Mathebula-Modongo, Heather Alexander, Christopher Serumola, Alison D. Grant, Katherine Fielding.
- Visualization: Andrew F. Auld, Andrew D. Kerkhoff, Alison D. Grant, Katherine Fielding.
- Writing original draft: Andrew F. Auld.
- Writing review & editing: Andrew F. Auld, Andrew D. Kerkhoff, Yasmeen Hanifa, Robin Wood, Salome Charalambous, Yuliang Liu, Tefera Agizew, Anikie Mathoma, Rosanna Boyd, Anand Date, Ray W. Shiraishi, George Bicego, Unami Mathebula-Modongo, Heather Alexander, Christopher Serumola, Goabaone Rankgoane-Pono, Pontsho Pono, Alyssa Finlay, James C. Shepherd, Tedd V. Ellerbrock, Alison D. Grant, Katherine Fielding.

References

- 1. World Health Organization. TB and HIV, and other comorbidities—2018 report. [cited 2020 Jun 3]. Available from: https://www.who.int/tb/areas-of-work/tb-hiv/en/.
- Gupta RK, Lucas SB, Fielding KL, Lawn SD. Prevalence of tuberculosis in post-mortem studies of HIVinfected adults and children in resource-limited settings: a systematic review and meta-analysis. AIDS. 2015; 29(15):1987–2002. https://doi.org/10.1097/QAD.00000000000802 PMID: 26266773.
- Gupta A, Nadkarni G, Yang WT, Chandrasekhar A, Gupte N, Bisson GP, et al. Early Mortality in Adults Initiating Antiretroviral Therapy (ART) in Low- and Middle-Income Countries (LMIC): A Systematic Review and Meta-Analysis. PLoS ONE. 2011; 6(12):e28691. https://doi.org/10.1371/journal.pone. 0028691 PMID: 22220193.
- World Health Organization. Latent TB Infection: Updated and consolidated guidelines for programmatic management. [cited 30 Aug 2018]. Available from: http://apps.who.int/iris/bitstream/handle/10665/ 260233/9789241550239-eng.pdf?sequence=1.

- Drain PK, Bajema KL, Dowdy D, Dheda K, Naidoo K, Schumacher SG, et al. Incipient and Subclinical Tuberculosis: a Clinical Review of Early Stages and Progression of Infection. Clin Microbiol Rev. 2018; 31(4). https://doi.org/10.1128/CMR.00021-18 PMID: 30021818.
- Hamada Y, Lujan J, Schenkel K, Ford N, Getahun H. Sensitivity and specificity of WHO's recommended four-symptom screening rule for tuberculosis in people living with HIV: a systematic review and metaanalysis. Lancet HIV. 2018; 5(9):e515–e23. <u>https://doi.org/10.1016/S2352-3018(18)30137-1</u> PMID: 30139576.
- Ahmad Khan F, Verkuijl S, Parrish A, Chikwava F, Ntumy R, El-Sadr W, et al. Performance of symptom-based tuberculosis screening among people living with HIV: not as great as hoped. AIDS. 2014; 28 (10):1463–72. https://doi.org/10.1097/QAD.00000000000278 PMID: 24681417.
- Rangaka MX, Wilkinson RJ, Glynn JR, Boulle A, van Cutsem G, Goliath R, et al. Effect of antiretroviral therapy on the diagnostic accuracy of symptom screening for intensified tuberculosis case finding in a South African HIV clinic. Clin Infect Dis. 2012; 55(12):1698–706. https://doi.org/10.1093/cid/cis775 PMID: 22955441.
- Melgar M, Nichols C, Cavanaugh JS, Kirking HL, Surie D, Date A, et al. Tuberculosis Preventive Treatment Scale-Up Among Antiretroviral Therapy Patients—16 Countries Supported by the U.S. President's Emergency Plan for AIDS Relief, 2017–2019. MMWR Morb Mortal Wkly Rep. 2020; 69(12):329–34. https://doi.org/10.15585/mmwr.mm6912a3 PMID: 32214084.
- Harries AD, Schwoebel V, Monedero-Recuero I, Aung TK, Chadha S, Chiang CY, et al. Challenges and opportunities to prevent tuberculosis in people living with HIV in low-income countries. Int J Tuberc Lung Dis. 2019; 23(2):241–51. https://doi.org/10.5588/ijtld.18.0207 PMID: 30808459.
- Pathmanathan I, Ahmedov S, Pevzner E, Anyalechi G, Modi S, Kirking H, et al. TB preventive therapy for people living with HIV: key considerations for scale-up in resource-limited settings. Int J Tuberc Lung Dis. 2018; 22(6):596–605. https://doi.org/10.5588/ijtld.17.0758 PMID: 29862942.
- Lawn SD, Wilkinson RJ, Lipman MC, Wood R. Immune reconstitution and "unmasking" of tuberculosis during antiretroviral therapy. Am J Respir Crit Care Med. 2008; 177(7):680–5. <u>https://doi.org/10.1164/ rccm.200709-1311PP PMID: 18202347</u>.
- Oni T, Burke R, Tsekela R, Bangani N, Seldon R, Gideon HP, et al. High prevalence of subclinical tuberculosis in HIV-1-infected persons without advanced immunodeficiency: implications for TB screening. Thorax. 2011; 66(8):669–73. https://doi.org/10.1136/thx.2011.160168 PMID: 21632522.
- Lawn SD, Harries AD, Wood R. Strategies to reduce early morbidity and mortality in adults receiving antiretroviral therapy in resource-limited settings. Curr Opin HIV AIDS. 2010; 5(1):18–26. <u>https://doi.org/10.1097/COH.0b013e328333850f PMID: 20046144</u>.
- Auld AF, Agizew T, Pals S, Finlay A, Ndwapi N, Boyd R, et al. Implementation of a pragmatic, steppedwedge cluster randomized trial to evaluate impact of Botswana's Xpert MTB/RIF diagnostic algorithm on TB diagnostic sensitivity and early antiretroviral therapy mortality. BMC Infect Dis. 2016; 16(1):606. https://doi.org/10.1186/s12879-016-1905-4 PMID: 27782821.
- Auld AF, Agizew T, Mathoma A, Boyd R, Date A, Pals SL, et al. Effect of tuberculosis screening and retention interventions on early antiretroviral therapy mortality in Botswana: a stepped-wedge cluster randomized trial. BMC Med. 2020; 18(1):19. <u>https://doi.org/10.1186/s12916-019-1489-0</u> PMID: 32041583.
- Obermeyer Z, Emanuel EJ. Predicting the Future—Big Data, Machine Learning, and Clinical Medicine. N Engl J Med. 2016; 375(13):1216–9. https://doi.org/10.1056/NEJMp1606181 PMID: 27682033.
- Hanifa Y, Fielding KL, Chihota VN, Adonis L, Charalambous S, Foster N, et al. A clinical scoring system to prioritise investigation for tuberculosis among adults attending HIV clinics in South Africa. PLoS ONE. 2017; 12(8):e0181519. https://doi.org/10.1371/journal.pone.0181519 PMID: 28771504.
- Grant AD, Charalambous S, Tlali M, Karat AS, Dorman SE, Hoffmann CJ, et al. Algorithm-guided empirical tuberculosis treatment for people with advanced HIV (TB Fast Track): an open-label, clusterrandomised trial. Lancet HIV. 2019. https://doi.org/10.1016/S2352-3018(19)30266-8 PMID: 31727580.
- Lawn SD, Bekker LG, Middelkoop K, Myer L, Wood R. Impact of HIV infection on the epidemiology of tuberculosis in a peri-urban community in South Africa: the need for age-specific interventions. Clin Infect Dis. 2006; 42(7):1040–7. https://doi.org/10.1086/501018 PMID: 16511773.
- WHO. Tuberculosis Country Profiles. [cited 2021 Apr 10]. Avaiable from: <u>https://www.who.int/tb/</u> country/data/profiles/en/.
- Lawn SD, Kerkhoff AD, Vogt M, Wood R. Diagnostic accuracy of a low-cost, urine antigen, point-of-care screening assay for HIV-associated pulmonary tuberculosis before antiretroviral therapy: a descriptive study. Lancet Infect Dis. 2012; 12(3):201–9. <u>https://doi.org/10.1016/S1473-3099(11)70251-1</u> PMID: 22015305.
- 23. Dryden-Peterson S, Lockman S, Zash R, Lei Q, Chen JY, Souda S, et al. Initial programmatic implementation of WHO option B in Botswana associated with increased projected MTCT. J Acquir

Immune Defic Syndr. 2015; 68(3):245–9. https://doi.org/10.1097/QAI.00000000000482 PMID: 25501611.

- Altman DG, Vergouwe Y, Royston P, Moons KG. Prognosis and prognostic research: validating a prognostic model. BMJ. 2009; 338:b605. https://doi.org/10.1136/bmj.b605 PMID: 19477892.
- 25. Corbett EL, Bandason T, Cheung YB, Makamure B, Dauya E, Munyati SS, et al. Prevalent infectious tuberculosis in Harare, Zimbabwe: burden, risk factors and implications for control. Int J Tuberc Lung Dis. 2009; 13(10):1231–7. PMID: 19793427.
- Lawn SD, Kranzer K, Edwards DJ, McNally M, Bekker LG, Wood R. Tuberculosis during the first year of antiretroviral therapy in a South African cohort using an intensive pretreatment screening strategy. AIDS. 2010; 24(9):1323–8. https://doi.org/10.1097/QAD.0b013e3283390dd1 PMID: 20386425.
- Royston P, Moons KG, Altman DG, Vergouwe Y. Prognosis and prognostic research: Developing a prognostic model. BMJ. 2009; 338:b604. https://doi.org/10.1136/bmj.b604 PMID: 19336487.
- Kerkhoff AD, Wood R, Vogt M, Lawn SD. Predictive value of anemia for tuberculosis in HIV-infected patients in Sub-Saharan Africa: an indication for routine microbiological investigation using new rapid assays. J Acquir Immune Defic Syndr. 2014; 66(1):33–40. <u>https://doi.org/10.1097/QAI.</u> 000000000000091 PMID: 24346639.
- Auld AF, Mbofana F, Shiraishi RW, Alfredo C, Sanchez M, Ellerbrock TV, et al. Incidence and determinants of tuberculosis among adults initiating antiretroviral therapy—Mozambique, 2004–2008. PLoS ONE. 2013; 8(1):e54665. https://doi.org/10.1371/journal.pone.0054665 PMID: 23349948.
- Benard A, Tessier JF, Rambeloarisoa J, Bonnet F, Fossoux H, Neau D, et al. HIV infection and tobacco smoking behaviour: prospects for prevention? ANRS CO3 Aquitaine Cohort, 2002. Int J Tuberc Lung Dis. 2006; 10(4):378–83. PMID: 16602400.
- Nanta S, Kantipong P, Pathipvanich P, Ruengorn C, Tawichasri C, Patumanond J. Screening scheme development for active TB prediction among HIV-infected patients. Southeast Asian J Trop Med Public Health. 2011; 42(4):867–75. PMID: 22299469.
- 32. Balcha TT, Skogmar S, Sturegard E, Schon T, Winqvist N, Reepalu A, et al. A Clinical Scoring Algorithm for Determination of the Risk of Tuberculosis in HIV-Infected Adults: A Cohort Study Performed at Ethiopian Health Centers. Open Forum Infect Dis. 2014; 1(3):ofu095. <u>https://doi.org/10.1093/ofid/ofu095</u> PMID: 25734163.
- Chowdhury MZI, Turin TC. Variable selection strategies and its importance in clinical prediction modelling. Fam Med Community Health. 2020; 8(1):e000262. <u>https://doi.org/10.1136/fmch-2019-000262</u> PMID: 32148735.
- Mupfumi L, Moyo S, Molebatsi K, Thami PK, Anderson M, Mogashoa T, et al. Immunological nonresponse and low hemoglobin levels are predictors of incident tuberculosis among HIV-infected individuals on Truvada-based therapy in Botswana. PLoS ONE. 2018; 13(1):e0192030. <u>https://doi.org/10.1371/journal.pone.0192030 PMID: 29385208</u>.
- Azman AS, Lessler J, Luquero FJ, Bhuiyan TR, Khan AI, Chowdhury F, et al. Estimating cholera incidence with cross-sectional serology. Sci Transl Med. 2019; 11(480). <u>https://doi.org/10.1126/</u> scitranslmed.aau6242 PMID: 30787170.
- Weng SF, Reps J, Kai J, Garibaldi JM, Qureshi N. Can machine-learning improve cardiovascular risk prediction using routine clinical data? PLoS ONE. 2017; 12(4):e0174944. <u>https://doi.org/10.1371/</u> journal.pone.0174944 PMID: 28376093.
- Zhou V. Random Forests for Complete Beginners. [cited 2021 Apr 10]. Available from: https:// victorzhou.com/blog/intro-to-random-forests/.
- Moons KG, Altman DG, Vergouwe Y, Royston P. Prognosis and prognostic research: application and impact of prognostic models in clinical practice. BMJ. 2009; 338:b606. <u>https://doi.org/10.1136/bmj.b606</u> PMID: 19502216.
- Gupta-Wright A, Corbett EL, Wilson D, van Oosterhout JJ, Dheda K, Huerga H, et al. Risk score for predicting mortality including urine lipoarabinomannan detection in hospital inpatients with HIV-associated tuberculosis in sub-Saharan Africa: Derivation and external validation cohort study. PLoS Med. 2019; 16(4):e1002776. https://doi.org/10.1371/journal.pmed.1002776 PMID: 30951533.
- Williams R. Standardized Coefficients in Logistic Regression. [cited 2021 Apr 12]. Available from: https://www3ndedu/~rwilliam/stats3/L04pdf.
- World Health Organization. Haemoglobin concentrations for the diagnosis of anaemia and assessment of severity. [cited 2020 Jun 1]. Available from: https://www.who.int/vmnis/indicators/haemoglobin/en/.
- Del Bene VE. Clinical Methods: The History, Physical, and Laboratory Examinations. 3rd edition. Chapter 218Temperature. [cited 2019 Dec 31]. Available from: https://www.ncbi.nlm.nih.gov/books/NBK331/.
- 43. Boyles TH, Nduna M, Pitsi T, Scott L, Fox MP, Maartens G. A Clinical Prediction Score Including Trial of Antibiotics and C-Reactive Protein to Improve the Diagnosis of Tuberculosis in Ambulatory

People With HIV. Open Forum Infect Dis. 2020; 7(2):ofz543. https://doi.org/10.1093/ofid/ofz543 PMID: 32042850.

- Aunsborg JW, Honge BL, Jespersen S, Rudolf F, Medina C, Correira FG, et al. A clinical score has utility in tuberculosis case-finding among patients with HIV: A feasibility study from Bissau. Int J Infect Dis. 2020; 92S:S78–S84. https://doi.org/10.1016/j.ijid.2020.03.012 PMID: 32171950.
- 45. Rudolf F, Haraldsdottir TL, Mendes MS, Wagner AJ, Gomes VF, Aaby P, et al. Can tuberculosis case finding among health-care seeking adults be improved? Observations from Bissau. Int J Tuberc Lung Dis. 2014; 18(3):277–85. https://doi.org/10.5588/ijtld.13.0517 PMID: 24670561.
- 46. Baik Y, Rickman HM, Hanrahan CF, Mmolawa L, Kitonsa PJ, Sewelana T, et al. A clinical score for identifying active tuberculosis while awaiting microbiological results: Development and validation of a multivariable prediction model in sub-Saharan Africa. PLoS Med. 2020; 17(11):e1003420. https://doi.org/10. 1371/journal.pmed.1003420 PMID: 33170838.
- **47.** Nkrumah B, Nguah SB, Sarpong N, Dekker D, Idriss A, May J, et al. Hemoglobin estimation by the HemoCue(R) portable hemoglobin photometer in a resource poor setting. BMC Clin Pathol. 2011; 11:5. https://doi.org/10.1186/1472-6890-11-5 PMID: 21510885.
- Sanchis-Gomar F, Cortell-Ballester J, Pareja-Galeano H, Banfi G, Lippi G. Hemoglobin point-of-care testing: the HemoCue system. J Lab Autom. 2013; 18(3):198–205. <u>https://doi.org/10.1177/</u> 2211068212457560 PMID: 22961038.
- Medina Lara A, Mundy C, Kandulu J, Chisuwo L, Bates I. Evaluation and costs of different haemoglobin methods for use in district hospitals in Malawi. J Clin Pathol. 2005; 58(1):56–60. https://doi.org/10.1136/ jcp.2004.018366 PMID: 15623483.
- Crowley C, Montenegro-Bethancourt G, Solomons NW, Schumann K. Validity and correspondence of non-invasively determined hemoglobin concentrations by two trans-cutaneous digital measuring devices. Asia Pac J Clin Nutr. 2012; 21(2):191–200. PMID: 22507604.
- Gayat E, Aulagnier J, Matthieu E, Boisson M, Fischler M. Non-invasive measurement of hemoglobin: assessment of two different point-of-care technologies. PLoS ONE. 2012; 7(1):e30065. <u>https://doi.org/10.1371/journal.pone.0030065</u> PMID: 22238693.
- PATH. Noninvasive technology for anemia detection. Seattle, WA, USA: PATH. [cited 2020 Aug 21]. Available from: http://www.path.org/publications/files/TS_update_anemia.pdf.
- Kerkhoff AD, Wood R, Cobelens FG, Gupta-Wright A, Bekker LG, Lawn SD. The predictive value of current haemoglobin levels for incident tuberculosis and/or mortality during long-term antiretroviral therapy in South Africa: a cohort study. BMC Med. 2015; 13:70. <u>https://doi.org/10.1186/s12916-015-0320-9</u> PMID: 25889688.
- Kerkhoff AD, Meintjes G, Burton R, Vogt M, Wood R, Lawn SD. Relationship Between Blood Concentrations of Hepcidin and Anemia Severity, Mycobacterial Burden, and Mortality Among Patients With HIV-Associated Tuberculosis. J Infect Dis. 2015. https://doi.org/10.1093/infdis/jiv364 PMID: 26136467.
- 55. Drakesmith H, Prentice AM. Hepcidin and the iron-infection axis. Science. 2012; 338(6108):768–72. https://doi.org/10.1126/science.1224577 PMID: 23139325.
- 56. Shapiro AE, Hong T, Govere S, Thulare H, Moosa MY, Dorasamy A, et al. C-reactive protein as a screening test for HIV-associated pulmonary tuberculosis prior to antiretroviral therapy in South Africa. AIDS. 2018; 32(13):1811–20. https://doi.org/10.1097/QAD.000000000001902 PMID: 29847333.
- Yoon C, Semitala FC, Atuhumuza E, Katende J, Mwebe S, Asege L, et al. Point-of-care C-reactive protein-based tuberculosis screening for people living with HIV: a diagnostic accuracy study. Lancet Infect Dis. 2017; 17(12):1285–92. https://doi.org/10.1016/S1473-3099(17)30488-7 PMID: 28847636.
- Lawn SD, Kerkhoff AD. Serum C-reactive protein measurement for ruling out HIV-associated tuberculosis. Int J Tuberc Lung Dis. 2013; 17(2):283–4. https://doi.org/10.5588/ijtld.12.0803 PMID: 23317967.
- WHO. WHO operational handbook on tuberculosis. Module 2: systematic screening for tuberculosis disease. [cited 2021 Apr 12]. Available from: https://apps.who.int/iris/bitstream/handle/10665/340256/ 9789240022614-eng.pdf.
- Santos VS, Goletti D, Kontogianni K, Adams ER, Molina-Moya B, Dominguez J, et al. Acute phase proteins and IP-10 as triage tests for the diagnosis of tuberculosis: systematic review and meta-analysis. Clin Microbiol Infect. 2019; 25(2):169–77. https://doi.org/10.1016/j.cmi.2018.07.017 PMID: 30076971.
- Holmes CB, Sikazwe I, Sikombe K, Eshun-Wilson I, Czaicki N, Beres LK, et al. Estimated mortality on HIV treatment among active patients and patients lost to follow-up in 4 provinces of Zambia: Findings from a multistage sampling-based survey. PLoS Med. 2018; 15(1):e1002489. <u>https://doi.org/10.1371/journal.pmed.1002489</u> PMID: 29329301.
- Tate JP, Sterne JAC, Justice AC, Veterans Aging Cohort S, the Antiretroviral Therapy Cohort C. Albumin, white blood cell count, and body mass index improve discrimination of mortality in HIV-positive individuals. AIDS. 2019; 33(5):903–12. https://doi.org/10.1097/QAD.0000000002140 PMID: 30649058.

- Odekunle FF, Odekunle RO, Shankar S. Why sub-Saharan Africa lags in electronic health record adoption and possible strategies to increase its adoption in this region. Int J Health Sci (Qassim). 2017; 11 (4):59–64. PMID: 29085270.
- Akanbi MO, Ocheke AN, Agaba PA, Daniyam CA, Agaba EI, Okeke EN, et al. Use of Electronic Health Records in sub-Saharan Africa: Progress and challenges. J Med Trop. 2012; 14(1):1–6. PMID: 25243111.
- 65. Karat AS, Omar T, von Gottberg A, Tlali M, Chihota VN, Churchyard GJ, et al. Autopsy Prevalence of Tuberculosis and Other Potentially Treatable Infections among Adults with Advanced HIV Enrolled in Out-Patient Care in South Africa. PLoS ONE. 2016; 11(11):e0166158. <u>https://doi.org/10.1371/journal.pone.0166158</u> PMID: 27829072.
- 66. van der Heijden YF, Karim F, Mufamadi G, Zako L, Chinappa T, Shepherd BE, et al. Isoniazid-monoresistant tuberculosis is associated with poor treatment outcomes in Durban, South Africa. Int J Tuberc Lung Dis. 2017; 21(6):670–6. https://doi.org/10.5588/ijtld.16.0843 PMID: 28482962.
- Samandari T, Bishai D, Luteijn M, Mosimaneotsile B, Motsamai O, Postma M, et al. Costs and consequences of additional chest x-ray in a tuberculosis prevention program in Botswana. Am J Respir Crit Care Med. 2011; 183(8):1103–11. https://doi.org/10.1164/rccm.201004-06200C PMID: 21148723.
- McNairy ML, Abrams EJ, Rabkin M, El-Sadr WM. Clinical decision tools are needed to identify HIV-positive patients at high risk for poor outcomes after initiation of antiretroviral therapy. PLoS Med. 2017; 14 (4):e1002278. https://doi.org/10.1371/journal.pmed.1002278 PMID: 28419097.
- 69. Shapiro AE, Chakravorty R, Akande T, Lonnroth K, Golub JE. A systematic review of the number needed to screen to detect a case of active tuberculosis in different risk groups. [cited 2020 Jun 10]. Available from: https://www.who.int/tb/Review3NNS_case_active_TB_riskgroups.pdf?ua=1.
- Khoury MJ, Iademarco MF, Riley WT. Precision Public Health for the Era of Precision Medicine. Am J Prev Med. 2016; 50(3):398–401. https://doi.org/10.1016/j.amepre.2015.08.031 PMID: 26547538.
- Pathmanathan I, Pevzner E, Cavanaugh J, Nelson L. Addressing tuberculosis in differentiated care provision for people living with HIV. Bull World Health Organ. 2017; 95(1):3. https://doi.org/10.2471/BLT. 16.187021 PMID: 28053356.
- Shoptaw S, Goodman-Meza D, Landovitz RJ. Collective Call to Action for HIV/AIDS Community-Based Collaborative Science in the Era of COVID-19. AIDS Behav. 2020. https://doi.org/10.1007/s10461-020-02860-y PMID: 32300993.
- Auld AF, Fielding KL, Gupta-Wright A, Lawn SD. Xpert MTB/RIF—why the lack of morbidity and mortality impact in intervention trials? Trans R Soc Trop Med Hyg. 2016; 110(8):432–44. <u>https://doi.org/10. 1093/trstmh/trw056</u> PMID: 27638038.
- 74. Date A, Modi S. TB screening among people living with HIV/AIDS in resource-limited settings. J Acquir Immune Defic Syndr. 2015; 68 Suppl 3:S270–3. <u>https://doi.org/10.1097/QAI.00000000000485</u> PMID: 25768866.
- 75. Pooran A, Theron G, Zijenah L, Chanda D, Clowes P, Mwenge L, et al. Point of care Xpert MTB/RIF versus smear microscopy for tuberculosis diagnosis in southern African primary care clinics: a multicentre economic evaluation. Lancet Glob Health. 2019; 7(6):e798–e807. https://doi.org/10.1016/S2214-109X (19)30164-0 PMID: 31097281.
- 76. Griesel R, Stewart A, van der Plas H, Sikhondze W, Rangaka MX, Nicol MP, et al. Optimizing Tuberculosis Diagnosis in Human Immunodeficiency Virus-Infected Inpatients Meeting the Criteria of Seriously III in the World Health Organization Algorithm. Clin Infect Dis. 2018; 66(9):1419–26. <u>https://doi.org/10.1093/cid/cix988 PMID: 29126226</u>.
- 77. Firmin S. Seeing the Forest for the Trees: An Introduction to Random Forest. [cited 2020 Jun 3]. Available from: https://towardsdatascience.com/seeing-the-forest-for-the-trees-an-introduction-to-random-forest-41a24fc842ac.
- Chen JH, Asch SM. Machine Learning and Prediction in Medicine—Beyond the Peak of Inflated Expectations. N Engl J Med. 2017; 376(26):2507–9. <u>https://doi.org/10.1056/NEJMp1702071</u> PMID: 28657867.
- 79. Rajkomar A, Dean J, Kohane I. Machine Learning in Medicine. N Engl J Med. 2019; 380(14):1347–58. https://doi.org/10.1056/NEJMra1814259 PMID: 30943338.